Author's response to reviews

Title: Alcohol consumption in Estonia and Finland: Finbalt survey 1994-2006

Authors:

   Kersti Pärna (kersti.parna@ut.ee)
   Kaja Rahu (kaja.rahu@tai.ee)
   Satu Helakorpi (satu.helakorpi@thl.fi)
   Mare Tekkel (mare.tekkel@tai.ee)

Version: 8 Date: 18 May 2010

Author's response to reviews:

Dear Editors, 18. May 2010

I would like to let you know that in the final version of the manuscript only minimal changes in the tables were performed. The headings of the columns in the tables were moved to the better position.

Sincerely yours,

Kersti Pärna

corresponding author